Author's response to reviews

Title: Vasectomy surgical techniques: a systematic review

Authors:

Dr Michel Labrecque (michel.labrecque@mfa.ulaval.ca)
Caroline Dufresne (bubblegum82@hotmail.com)
Mark A Barone (mbarone@engenderhealth.org)
Karine Saint-Hilaire (karinesthilaire@hotmail.com)

Version: 3 Date: 20 Apr 2004

As requested we made the following changes in the manuscript:
Title: all unnecessary capitalisation was removed.
Funding: this information from the title page was moved to the Acknowledgements section.
Highlighting: highlighting was removed from your manuscript text.
Abstract: abstract was shorten to 350 words.
Abstracts: rename 'Conclusion' was renamed as 'Conclusions'.
References: all references were placed before the full stop.
References: We ensured that authors' initials are given after their surnames and that full page numbers are given. We also checked references for accuracy.